Author’s response to reviews

Title: Surgical complications in neuromuscular scoliosis operated with posterior-only approach using pedicle screw fixation.

Authors:

Hitesh N Modi (hnm7678@yahoo.co.in)
Seung-Woo Suh (spine@korea.ac.kr)
Jae-Hyuk Yang (gurospine@naver.com)
Jae-Woo Cho (modispine@yahoo.co.in)
Jae-Young Hong (gpstylep@hanmail.net)
SuryaUdai Singh (hnm7678@yahoo.co.in)
Sudeep Jain (sudeep_im@yahoo.co.in)

Version: 7 Date: 19 April 2009

Author’s response to reviews: see over
Authors’ Response to Reviewers’ Comments:

Title: Surgical complications in neuromuscular scoliosis operated with posterior-only approach using pedicle screw fixation.

Reviewer: George H. H Thompson

Reviewer's report:

Minor Essential Revisions

(1). Page 8. Lumbar lordosis is not expressed with a minus sign. The fact that use the word lordosis distinguishes it from thoracic kyphosis.

Answer: We have added minus sign in the bracket as shown in tables.

Discretionary Revisions

(2). Please make sure all your results in the text are consistent with your tables. Also check your spelling.

Answer: We have matched the results exactly as in the tables. We have corrected table 5 showing the percentage according to results. Spellings are also checked.